Author's response to reviews

Title: Patients’ perspectives of care for type 2 diabetes in Bangladesh - a qualitative study.

Authors:

Christopher P Lewis (christopher.lewis@doctors.org.uk)
Zafar Ullah (Hss1anuz@leeds.ac.uk)
James Newell (j.n.newell@leeds.ac.uk)

Version: 2 Date: 23 April 2013

Author's response to reviews: see over
To whom it may concern,

Please find attached, a resubmission of our research paper entitled “Patients’ perspectives of care for type 2 diabetes in Bangladesh – a qualitative study.” (MS: 4176898168513937)

The re-submission contains the following edits as per editorial board request.

1. The manuscript now incorporates a title page with all the details of each author included and the full postal address of the submitting author

2. The article adheres to RATS guidelines for reporting qualitative studies. This statement is now included in the body of the manuscript

3. A competing interests section has been inserted at the end of the manuscript

4. Authors contributions have also been added at the end of the manuscript as requested.

The authors would kindly request that the article remains for consideration in BMC public health rather than any of the other BMC publications currently in print

Many thanks for your help and support with our manuscript submission. We look forward to hearing from you soon.

Best Wishes

CPL, JN & ZUAN